Author's response to reviews

Title: Nutrient adequacy during weight loss interventions: a randomized study in women comparing the dietary intake in a meal replacement group with a traditional food group

Authors:

Judith M Ashley (jashley@unr.edu)
Holly Herzog (holzog@gmail.com)
Sharon Clodfelter (sclofelter@washoeCounty.us)
Vicki Bovee (vbovee@westersurgical.com)
Jon Schrage (jschrage@kumc.edu)
Chris Pritsos (pritsos@cabnr.unr.edu)

Version: 4 Date: 14 June 2007

Author's response to reviews:

Per email provided June 13, 2007
Table 6 follows Table 5
Paragraphs in Table 6 are in separate cells and rows
The table title are above the respective tables